Author's response to reviews

Title: Reversible pegvisomant-related lipohypertrophy in two acromegalic women, clinical and radiological evidence of recurrence at the new site of injection: a case series

Authors:

- Vincenzo Rochira (vincenzo.rochira@unimore.it)
- Lucia Zirilli (lucia.zirilli@unimore.it)
- Chiara Diazzi (chiaradiazzi@gmail.com)
- Stefania Romano (s.romano@ausl.mo.it)
- Cesare Carani (cesare.carani@unimore.it)

Version: 4 Date: 28 October 2011

Author's response to reviews: see over
To: Professor Michael Kidd AM,
Editor of the Journal of Medical Case Reports
c/o BioMed Central
236 Gray's Inn Road
London WC1X 8HB
United Kingdom
Phone:+44 (0) 20 3192 2009
Fax :+44 (0)20 3192 2010

Dear Professor Michael Kidd AM

Please find enclosed the revised version (Revision 2) of the manuscript entitled “Reversible pegvisomant-related lipo hypertrophy in two acromegalic women, clinical and radiological evidence of recurrence at the new site of injection: a case series” that we are re-submitting to the Journal of Medical Case Reports for publication as Case series.

Answers to the Deputy Editor

“The authors need to significantly improve the standard of English in their manuscript. Other than this, the only other comment I would make relates to their figure, they need to be separated into Fig 1, 2, 3 etc. also uploaded individually as opposed to being embedded in their manuscript.”

In line with the Deputy Editor’s comments, may we request that your manuscript be checked over by a native English speaker and the language revised. If this is not possible, or has already been done, you may need to use a professional copyediting service. It is important that the quality of language is of a high standard before the manuscript can be published.

We revised the English in our manuscript. A native English speaker further revised the language.

We divided the four panels of Figure 1 in four different figures.
All the changes are highlighted in the revised version (the text added is in blue; the text deleted is in red).

Please, do not hesitate to contact me if you have any problems in reading the manuscript or if you need additional information.

Best Regards

Vincenzo Rochira and co-Authors